\documentclass[11pt, twoside]{article}
\input{macros-course}

\showsolutionsfalse

%\usepackage{hyperref}

\begin{document}
\pset{1}{September 5, 2001}{The beginning of lecture, Monday, Sept. 10,
2001}

\emph{Remember to include a \textbf{collaboration statement} and your
\textbf{TA's name} on the first page of your solutions.}

\textbf{Background Reading:} Review the following sections of the texts:
Velleman, \textit{How to Prove It}, \S1.1--5, \S4.1, \S5.1--2; Rosen,
\textit{Discrete Mathematics and Its Applications}, \S 1.1--2, \S1.4--6.
This reading covers mathematical notation and concepts that we hope you
find familiar.

\remove{
topics:
truth tables, sets and set operations, functions: domain, range,
injection, surjection, bijection, composition.
}

\textbf{Reading:} Week 1 Lecture Notes.  Velleman, \S 2.1--2,
\S 3.1--3.  Rosen, \S 1.3, \S 3.1.

\remove{
topics: proof by contradiction and cases, quantifiers.
}

\input{pset1-probs.tex}

\end{document}
